Author's response to reviews

Title: Evaluation Of Splanchnic Oximetry, Doppler Flow Velocimetry In The Superior Mesenteric Artery And Feeding Tolerance In Very Low Birth Weight IUGR And Non-IUGR Infants Receiving Bolus Versus Continuous Enteral Nutrition

Authors:

Valentina Bozzetti (vbozzetti@hotmail.com)
Giuseppe Paterlini (g.paterlini@hsgerardo.org)
Valeria Meroni (vale.meroni@gmail.com)
Paola De Lorenzo (paola.delorenzo@unimib.it)
Maria Grazia Valsecchi (grazia.valsecchi@unimib.it)
Diego Gazzolo (dgazzolo@hotmail.com)
Frank Van Bel (F.vanBel@umcutrecht.nl)
Gerard HA Visser (g.h.a.visser@umcutrecht.nl)
Paolo E Tagliabue (p.tagliabue@hsgerardo.org)

Version: 2 Date: 23 May 2012

Author's response to reviews:

We format the abstract and the text of the manuscript according to BMC style guidelines and we provide all the changes previously requested by the Editors.